\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\newcommand{\texttildeapprox}{{\fontfamily{pcr}\selectfont\texttildelow}}

\begin{document}


\def\RunningAuthor{Anuradha P and Madhumitha S S}
\firstPage{1694}
\articleType{Original Article}
\receivedDate{05 Aug 2020}
\acceptedDate{11 Sep 2020}
\revisedDate{10 Sep 2020}
\journalVolume{2020, 11 (SPL4)}
\journalDoi{ijrps.v11iSPL4.4357}
\copyrightYear{2020}
\def\authorCount{2}
\def\affCount{2}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Awareness of colour blindness among nurses}
\author{Anuradha~P\textsuperscript{*}\textsuperscript{1}, 
            Madhumitha~S~S\textsuperscript{2}~\\[5pt]{\textsuperscript{1}Department of Ophthalmology, Saveetha Medical College, Thandalam, Chennai, Tamil Nadu, India}~\\{\textsuperscript{2}Saveetha Medical College, Thandalam,  Chennai, Tamil Nadu, India}}

\begin{abstract}
The primary aim of this study is to find the awareness of colour blindness among 100 nurses in a hospital. Colour blindness or colour vision deficiency is the decreased ability to see and distinguish colours. It is one of the most common inherited disorders in human beings. It can also be acquired. It occurs due to the disorder or defect in the development of one or more of the three sets of colour sensing cones of the eye. There is no cure for colour vision deficiency. This study is a hospital-based descriptive cross-sectional study. Ethical clearance for the study was obtained from the institutional review board. A structured questionnaire on awareness of colour blindness was prepared. This was given to 100 nurses in the hospital. The data obtained were analysed with the help of a statistician. The result was that the nurses knew what colour blindness is, but they did not know much about the aetiology, transmission, treatment of the disorder. The nurses should be made more aware of colour blindness.  
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Awareness,\newline Colour blindness,\newline Colour vision deficiency
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Anuradha~P~\\ Phone:\ 9597855584~\\  Email:\ tammu.angel@ymail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Colour vision is the ability to discriminate colours based on the wavelength of light. This is due to the photoreceptors in the retina, mainly the cones. Colour blindness is the decreased ability to see and distinguish colours. It is mostly between red, green and blue, which are the three primary colours. It is also called colour vision deficiency. The retina has two types of specialised cells the rods and the cones. The rods are sensitive to dim lights and can detect light and dark, whereas the cones can detect colours. The brain uses input from these cone cells for colour perception. Colour blindness is due to disorder in the cones of the eyes either due to its absence or in its working. The most common cause is due to the inheritance of genes responsible for colour blindness. These genes are present mainly on X-chromosomes. So it affects males more than females \unskip~\citep{880205:20277177}.  Sometimes it can also occur due to acquired causes. Specific chronic illness like multiple sclerosis, diabetes mellitus, macular degeneration, leukaemia, retinitis pigementosa, parkinsonism can develop colour blindness. In early parkinsonism, the patients experience tritanopia \unskip~\citep{880205:20277187}. Conditions like cataract and glaucoma can also cause colour vision deficiency \unskip~\citep{880205:20277183}. It can also be due to accidents that cause damage to the optic nerve or retina of the eye. Certain medications like indomethacin and ethambutol can also lead to this disorder \unskip~\citep{880205:20277185}. The subject can have monochromacy, dichromacy or trichromacy. The most common type is the red-green deficiency \unskip~\citep{880205:20277188}. This can become very challenging in daily life. They cannot be allowed to become a pilot, electrician, doctor, engineer, driving and other jobs wherein colour differentiation play an important role. Ishihara test is the most common diagnostic test done for colour blindness. It is the most widely used screening test worldwide for the detection of colour blindness. There is no complete cure for this disorder. Current management chiefly aims at counselling \unskip~\citep{880205:20277181}. Awareness of this disorder is fundamental to guide these patients to choose their career, especially among medical personnel. Hence this awareness study was taken up in our hospital to study the awareness of colour vision among nurses.  
    
\section{Materials and Methods}
The study is a community-based cross-sectional descriptive study involving 100 nurses in a tertiary health care centre, Saveetha Medical College and Hospital, Chennai. Ethical clearance for the study was obtained from the institutional review board (IRB). A structured questionnaire with 17 questions about aetiology, mode of transmission and management was prepared for this study. 

This questionnaire was validated by two experts in the field of ophthalmology. This was distributed to the nurses to evaluate the level of awareness about the colour blindness. This data was collected, entered into a computer and analysed with the help of a statistician. Then the level of awareness about colour blindness among the nurses was calculated. The age group and gender of the subjects were also correlated with the level of awareness on colour blindness. These data were presented with the help of pie charts and tables.
    
\section{\textbf{RESULTS}}
The study was conducted among 100 nurses. Their age distribution was as shown in Table~\ref{tw-444bb1827257} and Figure~\ref{f-d075d9413bd9}. Majority of the people involved in the study were in the age group 19 to 23 (60\%) followed by people in the age group 24 to 28 (32\%) as given in Table~\ref{tw-444bb1827257}, Figure~\ref{f-d075d9413bd9}. 


\begin{table*}[!htbp]
\caption{\boldmath {age distribution of the participants } }
\label{tw-444bb1827257}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1489\linewidth-2\tabcolsep}p{\dimexpr.24769999999999996\linewidth-2\tabcolsep}p{\dimexpr.26800000000000004\linewidth-2\tabcolsep}p{\dimexpr.3354\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Age & Frequency & Percentage\\
\tblmidrule 
1 &
  19-23 &
  60 &
  60\%\\
2 &
  24-28 &
  32 &
  32\%\\
3 &
  29-33 &
  3 &
  3\%\\
4 &
  34-38 &
  3 &
  3\%\\
5 &
  39-43 &
  2 &
  2\%\\
6 &
  44-48 &
  1 &
  1\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Gender distribution of participants} }
\label{tw-50f8cc6e5d04}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1354\linewidth-2\tabcolsep}p{\dimexpr.23880000000000003\linewidth-2\tabcolsep}p{\dimexpr.2185\linewidth-2\tabcolsep}p{\dimexpr.4073\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Gender & Frequency & Percentage\\
\tblmidrule 
1 &
  Males &
  10 &
  10\%\\
2 &
  Females &
  90 &
  90\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {what are the three primary colours? } }
\label{tw-eb1a885896c0}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.13089999999999998\linewidth-2\tabcolsep}p{\dimexpr.387\linewidth-2\tabcolsep}p{\dimexpr.2321\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & What are the 3 primary colours? & Frequency & Percentage\\
\tblmidrule 
1 &
  Correct response &
  23 &
  23\%\\
2 &
  Incorrect response &
  46 &
  46\%\\
3 &
  Not aware &
  31 &
  31\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Have you heard of the term colour blindness?} }
\label{tw-8e1fbaa86644}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.12629999999999997\linewidth-2\tabcolsep}p{\dimexpr.37370000000000005\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Have you heard of the term colour blindness? & Frequency & Percentage \\
\tblmidrule 
1 &
  Yes &
  70 &
  70\%\\
2 &
  No &
  30 &
  30\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Did you know it is the decreased ability to see Colours?} }
\label{tw-0dddf68dc7e6}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1242\linewidth-2\tabcolsep}p{\dimexpr.3758\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Did you know it is the reduced ability to see or distinguish colours? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  63 &
  63\%\\
2 &
  No &
  37 &
  37\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Can it exist in one eye alone?} }
\label{tw-586da96683fe}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1241\linewidth-2\tabcolsep}p{\dimexpr.4051\linewidth-2\tabcolsep}p{\dimexpr.2208\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Can it exist in one eye alone? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  40 &
  40\%\\
2 &
  No &
  34 &
  34\%\\
3 &
  Not aware &
  26 &
  26\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Do you know why it occurs?} }
\label{tw-53530357daad}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.16009999999999998\linewidth-2\tabcolsep}p{\dimexpr.31740000000000002\linewidth-2\tabcolsep}p{\dimexpr.2725\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Do you know why it occurs? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  33 &
  33\%\\
2 &
  No &
  67 &
  67\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {A common cause of it?} }
\label{tw-406733c3a8b9}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1556\linewidth-2\tabcolsep}p{\dimexpr.3892999999999999\linewidth-2\tabcolsep}p{\dimexpr.2051\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Most common cause? & Frequency & Percentage\\
\tblmidrule 
1 &
  Genetic &
  69 &
  69\%\\
2 &
  Infection &
  19 &
  19\%\\
3 &
  Not aware &
  12 &
  12\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Mode of transmission?} }
\label{tw-4512ba935a99}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.12639999999999999\linewidth-2\tabcolsep}p{\dimexpr.3803999999999999\linewidth-2\tabcolsep}p{\dimexpr.2432\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Mode of transmission? & Frequency & Percentage\\
\tblmidrule 
1 &
  Inheritance &
  51 &
  51\%\\
2 &
  Contact &
  32 &
  32\%\\
3 &
  Not aware &
  17 &
  17\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Can vit A deficiency cause it? } }
\label{tw-a66d0097e070}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1511\linewidth-2\tabcolsep}p{\dimexpr.4253\linewidth-2\tabcolsep}p{\dimexpr.21849999999999994\linewidth-2\tabcolsep}p{\dimexpr.20509999999999998\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no & Can vit a deficiency cause it? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  50 &
  50\%\\
2 &
  No &
  29 &
  29\%\\
3 &
  Not aware &
  21 &
  21\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Can any drug cause it? } }
\label{tw-3fdcee951399}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.13769999999999998\linewidth-2\tabcolsep}p{\dimexpr.36230000000000004\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Can any drug cause this? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  31 &
  31\%\\
2 &
  No &
  46 &
  46\%\\
3 &
  Not aware &
  23 &
  23\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {More common in?} }
\label{tw-50547f995387}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1399\linewidth-2\tabcolsep}p{\dimexpr.3376\linewidth-2\tabcolsep}p{\dimexpr.2725\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & More common in? & Frequency & Percentage\\
\tblmidrule 
1 &
  Males &
  45 &
  45\%\\
2 &
  Females &
  38 &
  38\%\\
3 &
  Not aware &
  17 &
  17\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Most common type?} }
\label{tw-efd608dafe3c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1331\linewidth-2\tabcolsep}p{\dimexpr.3737\linewidth-2\tabcolsep}p{\dimexpr.2432\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Most common type? & Frequency & Percentage\\
\tblmidrule 
1 &
  Red green &
  43 &
  43\%\\
2 &
  Red yellow &
  45 &
  45\%\\
3 &
  Not aware &
  12 &
  12\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Monochromacy means?} }
\label{tw-bbc7b694226e}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.122\linewidth-2\tabcolsep}p{\dimexpr.414\linewidth-2\tabcolsep}p{\dimexpr.214\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Monochromacy means? & Frequency & Percentage\\
\tblmidrule 
1 &
  Total colour blindness &
  29 &
  29\%\\
2 &
  Partial colour blindness &
  51 &
  51\%\\
3 &
  Not aware &
  20 &
  20\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Diagnosis is by?} }
\label{tw-51f9fc24aec8}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1421\linewidth-2\tabcolsep}p{\dimexpr.3242\linewidth-2\tabcolsep}p{\dimexpr.2837\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Diagnosis is by? & Frequency & Percentage\\
\tblmidrule 
1 &
  Ishihara chart &
  33 &
  33\%\\
2 &
  Visual acquity &
  59 &
  59\%\\
3 &
  Not aware &
  8 &
  8\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Are people with colour blindness not allowed to perform specific jobs? } }
\label{tw-c1cd9634e1ed}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.12190000000000001\linewidth-2\tabcolsep}p{\dimexpr.4141\linewidth-2\tabcolsep}p{\dimexpr.214\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Are people with colour blindness not allowed to perform certain jobs? & Frequency  & Percentage\\
\tblmidrule 
1 &
  Yes &
  46 &
  46\%\\
2 &
  No &
  37 &
  37\%\\
3 &
  Not aware &
  17 &
  17\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Are people with colour blindness allowed to drive vehicles?} }
\label{tw-fb40d5fc4d64}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1331\linewidth-2\tabcolsep}p{\dimexpr.41410000000000004\linewidth-2\tabcolsep}p{\dimexpr.22080000000000002\linewidth-2\tabcolsep}p{\dimexpr.232\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Are people with colour blindness allowed to drive vehicles? & Frequency & Percentage\\
\tblmidrule 
1 &
  No &
  64 &
  64\%\\
2 &
  Yes &
  23 &
  23\%\\
3 &
  Not aware &
  13 &
  13\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Can it be completely cured?} }
\label{tw-c9f9567d218d}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.12850000000000001\linewidth-2\tabcolsep}p{\dimexpr.37150000000000006\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Can it be completely cured? & Frequency & Percentage\\
\tblmidrule 
1 &
  No &
  55 &
  55\%\\
2 &
  Yes &
  39 &
  39\%\\
3 &
  Not aware &
  6 &
  6\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Do you know anyone who is colour blind?} }
\label{tw-f1ddde1540b8}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1331\linewidth-2\tabcolsep}p{\dimexpr.3961\linewidth-2\tabcolsep}p{\dimexpr.2208\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.no. & Do you know anyone who is colour blind? & Frequency & Percentage\\
\tblmidrule 
1 &
  Yes &
  30 &
  30\%\\
2 &
  No &
  70 &
  70\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/e9396ed3-1e5a-485b-a961-0b1348b44fc6-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e9396ed3-1e5a-485b-a961-0b1348b44fc6-upicture1.png}{\includegraphics{images/e9396ed3-1e5a-485b-a961-0b1348b44fc6-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Age distribution of the participants}}
\label{f-d075d9413bd9}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/5dd00763-6275-4f26-aa86-31c833f07219-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5dd00763-6275-4f26-aa86-31c833f07219-upicture2.png}{\includegraphics{images/5dd00763-6275-4f26-aa86-31c833f07219-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Gender distribution of participants }}
\label{f-90a416be7cf8}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/1a4793a2-2550-4be1-bd4c-2130804ab003-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/1a4793a2-2550-4be1-bd4c-2130804ab003-upicture3.png}{\includegraphics{images/1a4793a2-2550-4be1-bd4c-2130804ab003-upicture3.png}}{}
\makeatother 
\caption{\boldmath {What are the three primary colours?}}
\label{f-e8856407e588}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/089586a9-800e-4f79-99dc-226f6d22b776-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/089586a9-800e-4f79-99dc-226f6d22b776-upicture4.png}{\includegraphics{images/089586a9-800e-4f79-99dc-226f6d22b776-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Have you heard of the term colour blindness?}}
\label{f-42c28e851a4a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6f5dee32-dcc9-411f-9662-ca034c3287c2-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6f5dee32-dcc9-411f-9662-ca034c3287c2-upicture5.png}{\includegraphics{images/6f5dee32-dcc9-411f-9662-ca034c3287c2-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Did you know it is the decreased ability to see Colours?}}
\label{f-75155c41ea02}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d6a6fb0a-0459-4a51-8da3-5fc6f6ae5d74-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d6a6fb0a-0459-4a51-8da3-5fc6f6ae5d74-upicture6.png}{\includegraphics{images/d6a6fb0a-0459-4a51-8da3-5fc6f6ae5d74-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Can it exist in one eye alone?}}
\label{f-0e49ac60236b}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7a7a7e2f-a678-4f7f-b4e2-e9de0e4d7cc0-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7a7a7e2f-a678-4f7f-b4e2-e9de0e4d7cc0-upicture7.png}{\includegraphics{images/7a7a7e2f-a678-4f7f-b4e2-e9de0e4d7cc0-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Do you know why it occurs?}}
\label{f-c8b839b575fc}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/405ca185-3db1-4574-8d6c-058a2334536f-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/405ca185-3db1-4574-8d6c-058a2334536f-upicture8.png}{\includegraphics{images/405ca185-3db1-4574-8d6c-058a2334536f-upicture8.png}}{}
\makeatother 
\caption{\boldmath {A common cause of it?}}
\label{f-326a5ea02a5e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/3ebfddee-6e69-4ac4-a2bc-49647dae7bab-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/3ebfddee-6e69-4ac4-a2bc-49647dae7bab-upicture9.png}{\includegraphics{images/3ebfddee-6e69-4ac4-a2bc-49647dae7bab-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Mode of transmission}}
\label{f-696afc94993d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/0b1cd800-ceb4-43ef-b87e-32eb3f1ef19c-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/0b1cd800-ceb4-43ef-b87e-32eb3f1ef19c-upicture10.png}{\includegraphics{images/0b1cd800-ceb4-43ef-b87e-32eb3f1ef19c-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Can vit A deficiency cause it? }}
\label{f-6652051dd2c3}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/65680b5c-1f4e-4414-b88c-492f02cb9bc7-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/65680b5c-1f4e-4414-b88c-492f02cb9bc7-upicture11.png}{\includegraphics{images/65680b5c-1f4e-4414-b88c-492f02cb9bc7-upicture11.png}}{}
\makeatother 
\caption{\boldmath {Can any drug cause it? }}
\label{f-38eff0651bed}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/367aeb8d-af1b-4a32-a039-945d3d096739-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/367aeb8d-af1b-4a32-a039-945d3d096739-upicture12.png}{\includegraphics{images/367aeb8d-af1b-4a32-a039-945d3d096739-upicture12.png}}{}
\makeatother 
\caption{\boldmath {More common in?}}
\label{f-206e37a62bec}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/40ef7764-1483-4791-ac8a-2a0246ec2bb2-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/40ef7764-1483-4791-ac8a-2a0246ec2bb2-upicture13.png}{\includegraphics{images/40ef7764-1483-4791-ac8a-2a0246ec2bb2-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Most common type?}}
\label{f-d99dff73914c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4f8a5152-dc0e-44b1-a457-b1f09a6160f7-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4f8a5152-dc0e-44b1-a457-b1f09a6160f7-upicture14.png}{\includegraphics{images/4f8a5152-dc0e-44b1-a457-b1f09a6160f7-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Monochromacy means?}}
\label{f-9bbc418ff3aa}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6a5528af-a087-45e8-997d-e4bf2018701b-upicture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6a5528af-a087-45e8-997d-e4bf2018701b-upicture15.png}{\includegraphics{images/6a5528af-a087-45e8-997d-e4bf2018701b-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Diagnosis is by? }}
\label{f-41ea1552d736}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8d4afa2e-9f35-4e60-8322-7fb4d8e0917a-upicture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8d4afa2e-9f35-4e60-8322-7fb4d8e0917a-upicture16.png}{\includegraphics{images/8d4afa2e-9f35-4e60-8322-7fb4d8e0917a-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Are people with colour blindness not allowed to perform specific jobs? }}
\label{f-45cbe51bbeaf}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/dcc357c3-ca01-4277-9424-d8e7a02da5f6-upicture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/dcc357c3-ca01-4277-9424-d8e7a02da5f6-upicture17.png}{\includegraphics{images/dcc357c3-ca01-4277-9424-d8e7a02da5f6-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Are people with colour blindness not allowed to perform specific jobs? }}
\label{f-e564434fdba8}
\end{figure}
\egroup
The gender distribution of people involved in the study was as in Table~\ref{tw-50f8cc6e5d04}, Figure~\ref{f-90a416be7cf8}. 90\% of nurses involved in the study were females, as shown in Table~\ref{tw-50f8cc6e5d04}, Figure~\ref{f-90a416be7cf8}. This may be due to the female preponderance in the occupation. The questionnaire had 17 questions. The response to each question was as shown in Table~\ref{tw-eb1a885896c0} and Figure~\ref{f-e8856407e588}. Only 23\% of the people were aware of the primary colours. 46\% answered it incorrectly, and 31\% were unaware as in the Table~\ref{tw-eb1a885896c0}, Figure~\ref{f-e8856407e588}. As shown in the Table~\ref{tw-8e1fbaa86644}, Figure~\ref{f-42c28e851a4a} 70\% of people who undertook the study have heard of the term colour blindness. 

63\% of people who took part in the study knows the actual meaning of colour blindness. 37\% of people are unaware of it as in the Table~\ref{tw-0dddf68dc7e6}, Figure~\ref{f-75155c41ea02}. When questioned if colour blindness can be unilateral 40\% answered it right, 34\% is incorrect, 26\% are not aware as shown in the Table~\ref{tw-586da96683fe}, Figure~\ref{f-0e49ac60236b}. Only 33\% are aware of the mechanism. 67\% are unaware as given in the Table~\ref{tw-53530357daad}, Figure~\ref{f-c8b839b575fc}. As in Table~\ref{tw-406733c3a8b9}, Figure~\ref{f-326a5ea02a5e} 69\% of people are aware that the common cause of it is genetic. 19\% have answered it incorrectly, and 12\% are not aware. 51\% of people are aware that the disease is inherited. 32\% of people answered it incorrectly, and 17\% are unaware as shown in the Table~\ref{tw-4512ba935a99}, Figure~\ref{f-696afc94993d}. Only 50\% of the people are aware that it can occur due to vitamin A deficiency as depicted in the Table~\ref{tw-a66d0097e070}, Figure~\ref{f-6652051dd2c3}. Only 31\% are aware that it can occur due to the intake of a particular drug as in the Table~\ref{tw-3fdcee951399}, Figure~\ref{f-38eff0651bed}. 45\% of people have answered it correctly that it is more common in males. 38\% of people have answered it incorrectly that it occurs in females. 17\% of people are not aware as shown in the Table~\ref{tw-50547f995387}, Figure~\ref{f-206e37a62bec}. 43\% of people know that red, green colour blindness is the most common type of colour blindness. 45\% answered it incorrectly. 12\% are not aware as given in the Figure~\ref{f-d99dff73914c}, Table~\ref{tw-efd608dafe3c}. As given in the Table~\ref{tw-bbc7b694226e}, Figure~\ref{f-9bbc418ff3aa} Only 29\% are aware of the term monochromacy. 51\% answered it incorrectly, and 20\% are unaware. 33\% are aware that the diagnosis is made using Ishihara chart. 59\% have answered it incorrectly, and 8\% are not aware as shown in the Table~\ref{tw-51f9fc24aec8}, Figure~\ref{f-41ea1552d736}. 46\% of people are aware that they are not allowed to engage in specific jobs. 37\% of people answered it incorrectly 17 \% are unaware as depicted in the Table~\ref{tw-c1cd9634e1ed}, Figure~\ref{f-45cbe51bbeaf}. As in Figure~\ref{f-e564434fdba8}, Table~\ref{tw-fb40d5fc4d64} 64 \% of people are aware that colour blind people cannot be allowed to drive vehicles. 23\% have said that they can be allowed to drive vehicles and 13\% are not aware. 55\% are aware that colour blindness cannot be cured entirely. 39\% have answered that it can be wholly cured and 6\% are not aware as given in the Table~\ref{tw-c9f9567d218d}, Figure~\ref{f-38ac08e2a5d7}. From the Table~\ref{tw-f1ddde1540b8}, Figure~\ref{f-e0a853aa984f} this we come to know that only 30 \% of people involved in our study have seen or known someone who is colour blind, 70\% of people do not know someone who is colour blind. 
    
\section{\textbf{DISCUSSION}}
Awareness of colour blindness among people, especially in people belonging to the health sector, can be very crucial. In this study conducted among 100 nurses, 70\% have heard of the term colour blindness, and 63\% knows what colour blindness is, whereas in an awareness study conducted among secondary school children in Emohua Local Government Area of Rivers State only 23.4\% were aware \unskip~\citep{880205:20277178}. Unilateral colour blindness is a rare condition wherein one eye is colour blind, other is normal. About 40\% of nurses who took part in this study are aware of it. In this study, when asked about the most common cause of colour blindness, 69\% answered it right as genetic. A study by kohl s et al. state that congenital total colour blindness is due to the mutation in the gene encoding alpha-subunit of cone photoreceptor cGMP gated channel \unskip~\citep{880205:20277184}. Vitamin A deficiency can also lead to colour blindness, and this was known to only 50\% of people who attended this study. A study on a middle-aged colour blind woman who had vitamin B12 deficiency had improvement in the colour vision on treatment with vitamin B12 injection. The study also stated that the vitamin B12 deficiency damages the optic nerve, which in turn results in visual problem \unskip~\citep{880205:20277176}. Colour vision deficiency can also be acquired due to the use of certain drugs like sildenafil, digoxin, ethambutol \unskip~\citep{880205:20277182}. 


\bgroup
\fixFloatSize{images/f0ac172c-1fd9-46d1-85f6-196a72be7e42-upicture18.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f0ac172c-1fd9-46d1-85f6-196a72be7e42-upicture18.png}{\includegraphics{images/f0ac172c-1fd9-46d1-85f6-196a72be7e42-upicture18.png}}{}
\makeatother 
\caption{\boldmath {Can it be completely cured?}}
\label{f-38ac08e2a5d7}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8d910e3c-cf04-46f8-975d-cbbb9389ae7c-upicture19.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8d910e3c-cf04-46f8-975d-cbbb9389ae7c-upicture19.png}{\includegraphics{images/8d910e3c-cf04-46f8-975d-cbbb9389ae7c-upicture19.png}}{}
\makeatother 
\caption{\boldmath {Do you know anyone who is colour blind?}}
\label{f-e0a853aa984f}
\end{figure}
\egroup
In this study, only 31 out of the 100 nurses knew that a drug could cause colour blindness. The preponderance of colour blindness in males and of red, green type was known by 45\%, 43\% respectively. A study conducted on school children also revealed a male, red, green colour deficiency preponderance \unskip~\citep{880205:20277186}. Using the Ishihara plates is the easiest way to diagnose colour blindness \unskip~\citep{880205:20277179}. Amongst the nurses who participated in this study, only 33\% knew about the diagnostic method for colour blindness. About 64\% of nurses knew that the colour blind people could not be allowed to drive vehicles. Colour-blind individuals cannot be allowed to drive vehicles as the perception of traffic signals, and warning signs are difficult for them. These people are also not allowed to become pilots, nurses, doctors, electricians, engineers, i.e., wherever the perception of colours is essential. Congenital colour blindness cannot be cured entirely, and this was known by 55 \% of nurses. A recent study by wenzel.k suggests that regular wearing of glasses for colour blindness can gradually improve the colour vision in them by the learning process, not by bringing change in the cones \unskip~\citep{880205:20277180}. This disorder affects about 1 in 12 men, 1 in 200 women.
    
\section{Conclusion}
It can be concluded that the nurses are not fully aware of colour blindness. Most of them know what colour blindness is and the most common aetiology of it. But they are not aware of the mechanism, challenges that it causes in daily life, the occupations which they cannot be employed in, diagnostic method and treatment. So more efforts should be taken, and nurses should be taught well about this disorder and how a colour blind person should be dealt with.

\textbf{Funding support}

The authors with this declare that they have no funding support for this study.

\textbf{Conflict of interest}

The authors now declare that they have no conflict of interest for this study.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
